

Aalborg Universitet

Patterns and characteristics of polypharmacy among elderly residents in Danish nursing homes

Astorp, Jacob; Giela, Mimoza; Jensen, Pernille; Bak, Rasmus D; Gazerani, Parisa

Published in: **Future Science OA**

DOI (link to publication from Publisher): 10.2144/fsoa-2020-0039

Creative Commons License CC BY 4.0

Publication date: 2020

Document Version Publisher's PDF, also known as Version of record

Link to publication from Aalborg University

Citation for published version (APA): Astorp, J., Gjela, M., Jensen, P., Bak, R. D., & Gazerani, P. (2020). Patterns and characteristics of polypharmacy among elderly residents in Danish nursing homes. *Future Science OA*, *6*(8), [FSO590]. https://doi.org/10.2144/fsoa-2020-0039

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- ? Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
 ? You may not further distribute the material or use it for any profit-making activity or commercial gain
 ? You may freely distribute the URL identifying the publication in the public portal ?

Take down policy

If you believe that this document breaches copyright please contact us at vbn@aub.aau.dk providing details, and we will remove access to the work immediately and investigate your claim.

For reprint orders, please contact: reprints@future-science.com

Patterns and characteristics of polypharmacy among elderly residents in Danish nursing homes

Jacob Astorp^{‡,1}, Mimoza Gjela^{‡,1}, Pernille Jensen^{‡,1}, Rasmus D Bak^{‡,1} & Parisa Gazerani^{*,1}^[D]

*Author for correspondence: gazerani@hst.aau.dk

[‡]Authors contributed equally

Aim: To identify patterns and characteristics of polypharmacy among elderly residents in Danish nursing homes in the Northern region of Denmark. **Materials & methods:** Twenty-five nursing homes were contacted, where each supplied 20 randomly selected anonymized residents' information. Residents were 65 years or older, concurrently taking five or more medications. Drug–drug interactions and potential adverse effects were investigated. **Results:** One hundred residents (68% females; 32% males) were included. The most prevalent co-morbid condition was cardiovascular disease, and the most prevalent medications were for gastrointestinal- and metabolism-related conditions. Age influenced the number of drugs (p = 0.013) and drug–drug interactions per resident (p = 0.039), with a positive correlation. **Conclusion:** Elderly residents of the studied nursing homes were potentially affected by an inappropriate polypharmacy.

Lay abstract: Multimorbidity is common among elderly and a leading factor for polypharmacy. We conducted this study on 100 Danish elderly residents (>65 years, on \geq 5 medications) in 25 nursing homes, anonymously and randomly chosen in Northern region of Denmark, to find characteristics of potential inappropriate polypharmacy. We found that age influenced the number of drugs (p = 0.013) and number of drug–drug interactions per resident (p = 0.039), but no sex-related difference was evident. Positive correlations were present between the number of drugs and drug–drug interactions. Elderly residents of the studied nursing homes were potentially affected by an inappropriate polypharmacy.

First draft submitted: 13 March 2020; Accepted for publication: 1 May 2020; Published online: 29 May 2020

Keywords: drug • drug-drug interaction • elderly • multimorbidity • nursing homes • polypharmacy

The percentage of the population above the age of 60 years old has increased drastically throughout the last few decades. It is estimated that a twofold increase of the world's elderly population will occur by 2050, reaching nearly 2.1 billion people aged 60 or above [1]. Age has been demonstrated to have an evident association with multimorbidity [2], which is the coexistence of multiple long-term health conditions in a person [3]. It is estimated that up to 95% of the population aged 65 years or older are affected by multimorbidity [3]. In a study performed in a Swedish nursing home, residents exhibited on average 16.8 ± 5.3 separate chronic health problems [4]. The top ten multimorbidities have been reported as hypertension, Type 2 diabetes, depression, lipid disorders, chronic obstructive pulmonary disease, ischemic heart disease, atrial fibrillation, obesity, asthma and dementia [5]. The most common affected organ systems have been identified as cardiovascular system followed by the central nervous system (CNS). Accordingly, the three most commonly prescribed drugs have been recorded as psycholeptics, cardiac medications and diuretics [6].

Multimorbidity results in an increased risk of hospitalization, loss of physical functioning, depression, premature death and polypharmacy [3,7,8]. No consensus currently exists for definition of polypharmacy. A systematic review of 110 studies on polypharmacy has stated that the most commonly used definition is based on concomitant consumption of five or more (>5) drugs [8], which has been reported in 44% of individuals aged 65 years or



Future Science

above [9]. Polypharmacy, is associated with higher risk of adverse outcomes such as frailty, disability, mortality and falls [10]. In one hand, polypharmacy seems necessary to manage multimorbidity in the elderly to target pathogeneses of a diverse range of diseases [11]. On the other hand, risks may arise when inappropriate polypharmacy exists, which is defined as an irrational prescription of too many drugs [12]. Appropriate polypharmacy occurs when the decisions concerning the prescription of multiple drugs to one patient rely on evidence-based practices, achieving the desired clinical effect without compromising the patient's quality of life by acquiring a reduction in drug toxicity [13]. However, inappropriate polypharmacy occurs when the prescription of multiple drugs to one patient fails to achieve the desired therapeutic effect, only resulting in increased risks and harmful health outcomes [13].

Inappropriate polypharmacy is a worldwide problem [12,14], including Scandinavian countries. For instance, Nyborg et al. [15] investigated use of potentially inappropriate medications (PIMs) among elderly living in nursing homes in Norway and demonstrated that 43.8% of the residents were prescribed at least one PIM, while 9.9% were treated with at least three PIMs. The prevalence of PIMs was also higher among females [15]. Inappropriate polypharmacy results in an increased risk of avoidable negative health outcomes [13] and has been associated with higher risks of adverse drug events (ADEs) [16], drug interactions, for example, drug-drug interactions (DDIs) [17], medication nonadherence and functional decline [18]. A subcategory of ADEs are adverse drug reactions (ADRs), defined as harmful or unpleasant reactions related to use of drugs [19]. Hanlon et al. [20] determined that polypharmacy consistently increases the risk of ADRs in elderly patients. In particular, it has been reported that this issue is higher in nursing homes [13], and patients on nine or more drugs are more than twice as likely to experience an ADR than those receiving less than nine [21]. This poses a serious problem because ADRs are difficult to distinguish from symptoms that are already present in the elderly, such as delirium and falls [22], making it difficult to choose the correct treatment, which can possibly lead to further inappropriate prescriptions [14]. Furthermore, the risk of hospitalization in the elderly with polypharmacy rises through an increase in ADEs [23]. Managing polypharmacy in the elderly, therefore, can be very challenging [22] and time consuming [24]. It is because of age-related changes in the physiology of the elderly population that cause alterations in drug disposition and metabolism [25-27], one example is the decline in the capacity of Phase I reaction enzymes (CYP450) [28]. Doan et al. [29] determined that 80% of the hospitalized elderly population with polypharmacy have a potential CYP450-mediated DDI, with the prevalence of potential DDIs increasing with the number of prescriptions.

Collectively, the higher prevalence of multimorbidity and polypharmacy, combined with age-related alterations in pharmacokinetics and pharmacodynamics, makes the older population more susceptible to ADEs [30]. Polypharmacy is also associated with anxiety or excitability, sleep difficulties, discomfort, weakness, confusion, tremors, hallucinations and feeling dizzy, demonstrating a negative impact on quality of life among elderly population [31]. Besides patient's safety, inappropriate polypharmacy poses an economic burden to healthcare system [32], due to hospital admissions or nonadherence to medications [14]. A Japanese cohort study highlighted that polypharmaceutical patients on one or more PIMs have a significantly higher hospitalization risk and more outpatient visit days, resulting in a 33% increase in healthcare cost when compared with polypharmaceutical patients on no PIMs [33]. Similar results have been found in Scandinavia, where a Swedish-based register study [34] determined that individuals on multiple medications (>5) account for 78.8% of total prescription medication acquisition costs. When compared with nonpolypharmaceutical patients, polypharmaceutical patients experienced increased prices per defined daily dose, with patients on ten or more prescribed drugs paying the most per dose [34]. Another Swedish study [35] has demonstrated that the increase in polypharmacy in recent years accounts for a large part of the increase in prescription drug expenditure. By comparing the prescription drug expenditure in 2005 and 2009, Hovstadius et al. [35] determined that the increase in polypharmacy between 2005 and 2009 could explain the entire increase in prescription drug expenditure in Sweden for that period. Several lists and guidelines have been developed around the world in attempt to prevent inappropriate polypharmacy and proper treatment of co-morbidities [36], including the US Beers criteria [15], Norwegian General Practice criteria [37], and Screening Tool of Older Patients' Prescriptions (STOPP) [38-40]. A modified version of the STOPP criteria is currently in use in Denmark [39,41], which has been developed by the Danish Geriatrics Society [42].

While some older studies have investigated patterns of polypharmacy in different regions in Denmark, including Funen [43] and Jutland [44], to the best of our knowledge, no recent Danish study has examined the pattern of polypharmacy and relationship between polypharmacy and DDIs, ADEs and ADRs, with a focus on nursing homes. Therefore, the aim of this study was to identify patterns and characteristics of polypharmacy in the elderly population living in nursing homes in Aalborg municipality, Northern region of Denmark, by investigating the DDIs and potential adverse effects. We proposed that our study could provide evidence for the existence of polypharmacy and potentially inappropriate polypharmacy in the elderly population above 65 years living in these nursing homes.

Methods

Study population

This study was conducted in 2019 based on medication lists obtained from nursing homes in Aalborg municipality, Northern Jutland, Denmark. The target population of this study was the polypharmaceutical elderly. Elderly was defined as being over 65 years of age, while polypharmacy was defined as the concurrent use of five or more prescription-based pharmaceutical dosage forms. In this definition, the time-limited medications, such as antibiotics, were not included, since the five or more medications should be taken regularly for a longer period of time. Furthermore, vitamins and food supplements that do not need a prescription (e.g., calcium, D vitamin, multivitamin) were not included in this category; hence, they were excluded from calculation of the total number of medications taken by each patient. Supplements that require a prescription, which in this study were vitamin B12 and potassium chloride, were included.

The participants in this study were recruited in the period of 7–27 November 2019, from nursing homes located in different parts of Aalborg municipality with following postal codes: 9000 Aalborg, 9200 Aalborg south west, 9210 Aalborg south east, 9220 Aalborg east, 9270 Klarup, 9280 Storvorde, 9310 Vodskov, 9382 Tylstrup and 9400 Noerresundby. The different nursing homes were identified using a publicly available list of nursing homes in Aalborg municipality [45].

Due to the nature of this study, the ethical approval from the North Denmark Region Committee on Health Research Ethics was not required. Since participants' information was provided anonymously by nursing homes, no consent form was required.

Study process

In total, 25 nursing homes were identified and contacted (total resident capacity = 1165) by personal attendance in order to determine their interest in participating in this study voluntarily. Each nursing home was asked to supply the following information from 20 randomly selected residents among those aged 65 years or older, who were concurrently taking five or more medications: age, sex, co-morbidities and medication list. Each nursing staff was asked to anonymize the data for each resident before turning it over in order to comply with the general data protection regulation, which is currently in effect in the European Union. Hence, the study remained anonymous. Recruitment process is depicted in Figure 1.

When illustrating the total resident capacity, it is important to state that this number does not reflect the actual number of residents at nursing homes. The capacity number has been reported instead of the actual number of current residents, since the number of residents is continuously changing.

The nursing homes that expressed interest were given a period of 1 week to consider whether or not they wanted to participate, followed by a period of 1 week to obtain the data. Afterward, the requested information was either collected from the nursing homes in person or received by E-mail. The nursing homes that refused to participate expressed that this was due to lack of time or nursing staff, participation in other projects/studies, or other reasons.

Data analysis

After receiving the datasets, data management was performed using Google sheet. The data were collected in three separate sheets, of which the first contained age, sex and co-morbidities; the second contained the individual medication lists; and the third contained the identified DDIs among the medications. In the first sheet, the co-morbidities were divided into following classes: 'cardiovascular,' 'central nervous system,' 'dermatological,' 'endocrinological,' 'eyes and ears,' 'gastrointestinal,' 'musculoskeletal,' 'pulmonal,' 'psychological' and 'urogenital.' The co-morbidities were indicated as numbers, which means a total number of co-morbidities was defined and recorded.

Prescription medications taken by each subject of the study were sorted in the second sheet according to their Anatomical Therapeutic Chemical (ATC) code, which has been the recommended classification system for drug utilization studies since 1981 by the WHO (Geneva, Switzerland) [46]. This categorization was performed at the second level of the ATC code, meaning that drugs were sorted according to their therapeutic subgroup (e.g., 'A01, stomatological preparations,' 'A02, drugs for acid-related disorders') [46].



Figure 1. Flowchart illustrating the study steps of recruitment of the residents' information.

The third sheet was completed utilizing two different drug interaction databases. One is run by the Danish Medication Agency [47] and the other by the US FDA [48]. Both databases were used to identify potential DDIs among the different medications. In the FDA database, the potential DDIs were divided into minor, moderate and major severities. The minor DDIs were clinically insignificant and could be used together, if a monitoring plan was made. The moderate DDIs were of modest clinical significance, these combinations should usually be avoided or only used under special circumstances. The major DDIs were highly clinically significant, whereby these combinations should be avoided [48]. In the Danish database, minor potential DDIs were those with clinical consequences known to be insignificant or nonexistent. Moderate potential DDIs were combinations of drugs that can be used therapeutically, if certain precautions are taken (e.g., dosage adjustments or changing the timing of ingestion). Major potential DDIs were combinations that are contraindicated or combinations that consisted of at least one drug with no scientifically proven effect [47]. In the database run by the FDA, Microlax (sodium lauryl sulfoacetate), Bromazepam, Flupentixol, Thiamazol, Zuclopenthixol, Moxonidin, Lercanidipine and Chlorprothixen were not available, due to the lack of approval in the USA. These drugs could not be checked for any potential interactions using the FDA database. After managing the data in Google sheets, different illustrations of data were made, including tables, graphs and pie charts.

Table 1. Demographic characteristics divided into four age groups. Values are expressed as number of subjects and

percentages.					
Characteristics	Age (years)				Total
	65–74	75–84	85–94	95+	
Population	14 (14%)	44 (44%)	36 (36%)	6 (6%)	100
Sex					
– Female	6 (8.8%)	32 (47.1%)	24 (35.3%)	6 (8.8%)	68
– Male	8 (25%)	12 (37.5%)	12 (37.5%)	0 (0%)	32
Co-morbidities					
- 0-4	4 (12.5%)	12 (37.5%)	12 (37.5%)	4 (12.5%)	32
- 5-9	10 (16.1%)	29 (46.8%)	21 (33.9%)	2 (3.2%)	62
- 10+	0 (0%)	3 (50%)	3 (50%)	0 (0%)	6
Drugs					
- 5-9	11 (16.2%)	23 (33.8%)	29 (42.6%)	5 (7.4%)	68
- 10+	3 (9.4%)	21 (65.6%)	7 (21.9%)	1 (3.1%)	32

Statistical analysis

The following statistical tests were performed using the software IBM SPSS Statistics version 25 for Mac OS X. A statistical significance level of 0.05 and a 95% CI were used to state whether a statistical significance was present. Prior to the statistical tests, all data (age, sex, number of drugs and number of DDIs) were tested for normal distribution by using the Shapiro–Wilk test.

Three nonparametric tests were performed. A Mann–Whitney U test was utilized to compare the number of drugs and the number of DDIs among the female and male residents and the Kruskal–Wallis test was performed to compare the number of drugs and the number of DDIs among four different age groups (65–74; 75–84; 85–94; 95+). The results from both the Mann–Whitney U and the Kruskal–Wallis tests are presented as median (25–75 percentiles). A Kendall's tau-b correlation was performed to investigate the relationship between the number of drugs and the DDIs between the two databases.

Results

Demographics

One hundred (n = 100) subjects from eight different nursing homes were identified as eligible for data analysis. The mean (\pm standard deviation) age in this population was 82.9 \pm 7.7 years and out of 100 subjects, 68% were females. This population presented on average 5.8 \pm 2.2 co-morbidities, taking on average 8.5 \pm 2.6 prescription-based pharmaceutical dosage forms. The demographics of these subjects are presented in Table 1.

Co-morbidities & medications

A total of 587 co-morbidities among the 100 residents were registered. As shown in Figure 2, the cardiovascular system was the most affected system with 157 identified co-morbidities. The second most affected system was the musculoskeletal system (86 co-morbidities), followed by the CNS (79 co-morbidities).

A total of 848 prescriptions were identified and categorized according to the ATC codes. Table 2 displays the sorting of these prescription drugs into their anatomical main groups.

Drug-drug interactions

Figure 3 illustrates the distribution of minor, moderate and major potential DDIs for both the FDA and Danish drug interaction databases. When inspecting the data from the Danish database (A), the minor DDIs made up the majority (68.6%) of the total number of interactions (n = 226). Comparatively, the moderate potential interactions were accountable for the majority (76.8%) of the total number of potential interactions (n = 786), when inspecting the data from the FDA database (B).

In Table 3, the ten most common DDIs were identified, of which the most common ones were Furosemide and Metoprolol (number of cases = 13). Among the ten most common DDIs, only DDIs of moderate severity were identified, which means no major DDIs were presented. Furthermore, the potential clinical effects of these DDIs are presented. The minor DDIs were not included.



Figure 2. The distribution of co-morbidities by affected systems. CNS: Central nervous system; GI: Gastrointestinal.

Table 2. The distribution of prescribed medications by the Anatomical Therapeutic Chemical codes.			
Overall drug classes	Number of prescriptions identified	Number of patients receiving drugs	Average no. of drugs received by each subject with 1≤ active prescription
Alimentary tract and metabolism (A)	240	92	2.6
Blood and blood-forming organs (B)	64	62	1.0
Cardiovascular system (C)	191	83	2.3
Dermatologicals (D)	14	12	1.2
Genitourinary system (G)	16	14	1.1
Systemic hormonal preparations (H)	15	15	1
Anti-infectives for systemic use (J)	8	8	1
Antineoplastic and immunomodulating agents (L)	5	5	1
Muscular-skeletal system (M)	27	27	1
CNS drugs (N)	219	83	2.6
Antiparasitic products (P)	2	2	1
Respiratory system (R)	18	15	1.2
Sensory organs (S)	24	13	1.8
Other drugs [†]	5	5	1

[†]Other drugs include categories of drugs, which only contained one prescription. These are: digestives including enzymes (A09), antianemic preparations (B03), antibiotics and chemotherapeutics for dermatological use (D06), antivirals (J05), anti-inflammatory and antirheumatic preparations (M01). CNS: Central nervous system.

Comparisons

The Shapiro–Wilk test of normality revealed no significance for co-morbidities, sex or DDIs in both the FDA and the Danish databases, whereas a significance was observed for age (p = 0.330). This means that data from sex, co-morbidities, number of drugs and DDIs did not follow a normal distribution, whereas the data from age were normally distributed.



Figure 3.	The distribution of potential drug-drug interactions found in the Danish drug interaction database (DK)
and the FD	DA drug interaction database (USA). The DDIs were divided into three categories based on severity: minor,
moderate	and major.
DDI: Drug-	-drug interaction.

Table 3. The ten most common moderate and major drug–drug interactions identified in the Danish and the FDA drug interaction databases. The potential clinical effects and the severity based on clinical significance are from the databases. The minor drug–drug interactions were not included.				
Rank	DDI	Number of cases	Potential clinical effect	Severity
1	Furosemide + Metoprolol	13	Hyperglycemia and hypertriglyceridemia	Moderate
2	Furosemide + Pantoprazole	11	Hypomagnesemia	Moderate
3	Furosemide + Sodium picosulfate	9	Fluid and electrolyte disturbances	Moderate
4	Metoprolol + Amlodipine	9	Further reductions in heart rate, cardiac conduction and cardiac contractility	Moderate
5	Clopidogrel + Pantoprazole	8	Reduced bioactivation of clopidogrel and the therapeutic efficacy	Moderate
6	Metformin + Insulin	7	Hypoglycemia	Moderate
7	Mirtazapine + Metoprolol	7	Hypotension	Moderate
8	Furosemide + Insulin	7	Hyperglycemia, glucose intolerance, new-onset diabetes mellitus and/or exacerbation of pre-existing diabetes	Moderate
9	Furosemide + Metformin	7	Increased metformin levels potentially leading to lactic acidosis	Moderate
10	Atorvastatin + Clopidogrel	6	Reduced bioactivation of clopidogrel and its antiplatelet effect	Moderate

Age

The results from the Kruskal–Wallis test comparing the association between number of drugs per resident and the different age groups are presented in Table 4. The results showed significant differences (p = 0.013) between the

Table 4. The median (25–75 percentiles) number of drugs and drug–drug interactions in the Danish and the FDA			
databases in the four different age groups.			
Age (years)	Number of drugs	DDI (Danish database)	DDI (FDA database)
65–74	8 (6.5–9.5)	1 (0–4)	6 (4–9.5)
75–84	9 (8–11)	2 (1–4)	7.5 (4–13.75)
85–94	7 (6–9)	1.5 (0.25–3)	5 (2.25–7)
95+	7 (6–9)	1 (1–1)	4 (1–7)
DDI: Drug-drug interaction.			

Table 5. The median (25–75 percentiles) number of drugs and drug–drug interactions in the Danish and the FDA database for females and males.

Sex	Number of drugs	DDI (Danish database)	DDI (FDA database)
Female	9 (7–10)	2 (1–3)	6 (3–12)
Male	7.5 (6–9.75)	1 (0.25–3)	6 (2.25–8)
DDI: Drug-drug interaction			

four different age groups. Furthermore, an investigation on the relation between the number of DDIs per resident and age groups was performed by using the Kruskal–Wallis test. As seen in Table 4, the highest number of DDIs per resident was observed in the age group of 75–84 years old. A statistical significance in the FDA database (p = 0.039) was revealed, however no significant difference was observed in the Danish database (p = 0.122).

Sex

As seen in Table 5, there was a higher number of drugs per resident among the females compared with the males, however, this was not significant (p = 0.070). Furthermore, a Mann–Whitney U test was performed to analyze if an association exists between the number of DDIs per resident and sex, the results of which are illustrated in Table 5. No significant difference was found in the number of DDIs per resident between females and males in the FDA (p = 0.269) or the Danish database (p = 0.397).

Correlations

Kendall's tau-b performed on the number of drugs per resident and age, revealed a correlation coefficient of -0.151, which indicates a minor inverse correlation between these two variables, suggesting that as age increases the number of drugs decreases. Kendall's tau-b was also performed on data including number of drugs and the number of DDIs per resident (both in the FDA and Danish database). The correlation coefficients were 0.400 between DDIs in the Danish database and the number of drugs, and 0.548 between DDIs in the FDA database and the number of drugs. These correlation coefficients indicate the presence of positive correlations between all variables, meaning that as one variable increases, the other also increases.

Discussion

This preliminary study provides the first evidence of potentially inappropriate polypharmacy existing in Danish elderly resident homes, located at Aalborg municipality. Findings from this study, therefore, must be viewed as a starting point of research concerning polypharmacy in Danish elderly resident homes.

Demographics

In the present study, 100 residents from nursing homes in Aalborg municipality were included, of which 68% were females. Females constitute 54% of the Danish elderly population [49]. The observed difference between the general elderly population and the sample population in this study could suggest that females are more susceptible to polypharmacy and thus inclusion in the study, which would be in accordance with previous findings [33,50,51]. However, one study of European nursing home residents found that 73% of a total study population of 4023 subjects were females [51]. This indicates that the high percentage of females is a reflection of the over-representation of females among nursing home residents in Europe. This might be attributed to sex-based differences in mortality, a difference that increases with age [52].

Furthermore, the mean age of the present population was 82.9 ± 7.7 years. This number is comparable to numbers found across Europe. One study determined an average age of 83.5 ± 9.4 years among nursing home residents across seven different European countries [51]. Akner's subjects also were 85 ± 7 years of age in a nursing home in Sweden, pointing out to similar results in Scandinavia [4].

Collectively, the demographics of the subjects included in this study seem representative of Danish nursing home residents in general. This is supported by a national study conducted by the Danish Ministry of Health (Sundhedsog Ældreministeriet) [53]. They demonstrated that on average, 68% of nursing home residents in Denmark were females. While a mean age for current residents of nursing homes was not reported, they did find that the average age for new residents in nursing homes was 83.7 years of age, with residents on average living in nursing homes for 2 years and 8 months, pointing out to a true mean slightly higher than 83.7 years.

Co-morbidities & medications

Among the residents, an average of 5.8 co-morbidities was identified by asking nursing home staff to provide information on co-morbidities present in any given subject, which they may have done in a variety of ways. This number is in contrast to the findings by Akner [4], who found an average of 16.8 ± 5.3 chronic health problems among 70 residents from a Swedish nursing home. While this difference could be due to a true difference in frequency of multimorbidity between nursing homes, it most likely reflects methodological differences. Akner [4] combined medical history, clinical examinations and patient history to assemble a more complete impression of the health status of the participants, leading to a higher sensitivity in detecting chronic diseases.

The present study identified that the most commonly affected system was the cardiovascular system, followed by the musculoskeletal system and the CNS. This is not in accordance with previous findings. Akner [4] identified that the most common co-morbidities were of neuropsychiatric, cardiovascular and gastrointestinal nature. These differences might be due to an over-representation of neuropsychiatric disorders, which were present in all 70 residents in his study [4].

The number of prescribed medications per resident in this study was 8.5 ± 2.6 , of which the three most commonly prescribed medications targeted the alimentary tract and metabolism, CNS and cardiovascular system. This is more than the 6.6 ± 3.2 prescribed continuously taken medications that Akner determined among the 70 residents in a Swedish nursing home [4]. This finding most likely stems from the inclusion criteria of more than five drugs in the present study skewing the average number of medications. However, it has been demonstrated that this cutoff of more than five drugs has predictable clinical discriminatory properties [10].

Another study [6] demonstrated that the three most commonly prescribed medications were psycholeptics, cardiac medications and diuretics, which are related to the systems identified as the most common affected systems in the present study. The classification of drugs according to the ATC codes conducted by Straand *et al.* [6] confirms this by identifying the CNS as the most commonly medicated system, followed by the cardiovascular system.

An inverse correlation between age and number of drugs was detected. This finding is counterintuitive, as it is well known in the literature that advanced age and polypharmacy are associated [9,27]. However, when dividing the population into four subgroups, the comparison of the of number of drugs demonstrated a statistically significant difference. The highest number of drugs per resident, based on medians, was among 75–84 year olds, while the lowest number of drugs per resident was seen among the elderly above the age of 85 years. These findings are in line with the results from the SHELTER study [51], which demonstrated fewer cases of excessive polypharmacy (>9 drugs) in the oldest nursing home residents. It has been proposed that this inverse correlation is caused by the nature of end-of-life care. When starting a new therapeutic regimen, remaining life expectancy is weighed against time required for therapeutic effect. The risk of ADRs is also higher in this patient group. These factors in combination can lead to fewer cases of excessive polypharmacy [3], as such it can be expected that patients living with polypharmacy have a reduced life span. This number could also reflect that patients on fewer drugs have fewer co-morbidities and consequently may live longer.

There was an obvious variation in the number of subjects and the distribution of females and males within the different age groups. The results may have been influenced by these differences between the groups. This makes it difficult to assign the observed association entirely on each age group.

Additionally, a higher number of coprescribed drugs was observed among females when compared with males. While this result was not statistically significant, the tendency is in accordance with other studies. Among others, one study by Venturini *et al.* [55] highlighted that the number of drugs was higher among females, compared with

males. It has been found that among Finnish and Norwegian elderly, women are slightly more likely to seek primary healthcare, which might partly explain the difference between the sexes [56]. Females also tend to report significantly more morbidity than their male counterparts, a difference that may, in part, be accounted for by gynecological and obstetrical diagnoses [57]. Accordingly, part of the difference between men and women can be attributed to the use of contraceptive pills and hormones for menopause [58]. While the sample included in this study was above 65 years, this use of contraceptives might mean that women are more accustomed to the use of drugs throughout their life and into old age [55].

Drug-drug interactions

In the present study, when the Danish interaction database was used, 226 potential DDIs were identified, of which most were with the nature of minor severity. In contrast, when the FDA database was used, 786 potential DDIs were identified, of which most were with the nature of moderate severity. This indicates that the FDA database had a higher sensitivity for detecting DDIs when compared with the Danish database. One explanation for the difference between number and severity of potential DDIs reported by the different DDI databases may be that these databases have different physiological foci. Interaktionsdatabasen.dk is based mainly on pharmacokinetic studies, while the interaction checker on Drugs.com is based on both pharmacodynamic and pharmacokinetic studies. For this reason, there might be some pharmacodynamic DDIs that are not included by the Interaktionsdatabasen.dk. Furthermore, Interaktionsdatabasen.dk is based on data from PubMed and Embase, while Drugs.com is based mainly on data from the American medical information suppliers. Therefore, while the sensitivity was higher when using Drugs.com, the fact that the data are based on extraction from a more heterogeneous population, raises questions about the specificity of the results when applied to the very homogenous Danish population in this study. In fact, it is a known problem that different DDI screening tools offer different sensitivity and specificity. Furthermore, it has also been reported that DDI databases under-report clinically relevant DDIs and in some cases, over report insignificant drug interactions [59,60].

Combining data from both databases, the most commonly identified DDIs were Furosemide and Metoprolol, followed by Furosemide and Pantoprazole. To the best of our knowledge, no comparable, recent studies have investigated patterns of DDIs in polypharmaceutical nursing home residents. This means that when comparing the results of the present study, it is important to note that other studies may have dissimilar study populations. The identified DDIs were not in accordance with the findings from the study by Sönnerstam *et al.* [61], who found the most common clinically relevant DDI to be Furosemide and Citalopram, followed by Acetylsalicylic acid and Citalopram.

The highest number of DDIs per resident was observed in the age group 75–84 years. No significant relation was found between age groups and DDIs in the Danish interaction database, whereas a significant difference in DDIs was determined between age groups in the FDA database. This could be due to the majority of the patients in this age group (75–84 years) taking a higher number of medications, which can increase the risk of DDIs. Goldberg *et al.* [62] highlighted that for patients taking two medications, the risk of DDIs was 13%, compared with 38% for patients taking five medications and 82% for patients taking seven medications or more.

Additionally, the correlation coefficient comparing the number DDIs per resident in the Danish database and the number of drugs, indicated a positive correlation (0.400), whereas the correlation coefficient from the comparison of the number of DDIs per resident in the FDA and the number of drugs indicated a stronger positive correlation (0.548). These correlations are in accordance with the results from the aforementioned study, which demonstrated an elevated risk of DDIs when increasing the number of drugs [62]. These findings could be further explained by the fact that the amount of potential pairwise interactions increases exponentially with each coprescribed drug [63].

No significant difference was observed in the total number of DDIs among males and females in the FDA or the Danish database. These findings are supported by the medians of DDIs in both males and females, which are nearly identical. When considering the literature, this finding is unexpected. A study by Venturini *et al.* [55] reported that females are more likely to take a higher number of medications than men, putting them at higher risk for potential DDIs. A summary of our findings compared with other studies is presented in Table 6.

Clinical consequences of DDIs

Information regarding documented clinical consequences for the residents included in this study was not available. Therefore, we reviewed theoretical and clinically observed adverse effects for the ten most commonly identified potential DDIs in the present study. Table 7 summarizes the findings.

Table 6. Summary of the present findings in comparison with previous studies in the literature.			
Our findings	Others findings		
68% of elderly at nursing homes with polypharmacy are females	58% of the population above 65 years living at nursing homes in Denmark are females [49]		
Mean (\pm SD) age = 82.9 \pm 7.7 years	In Europe [51], the mean age is 83.5 \pm 9.4 years In Sweden [4], the mean age is 85 \pm 7 years New resident average age in Denmark [64]: 83.7 years		
5.8 co-morbidities in average	Akner [4] found 16.8 \pm 5.3 in average		
8.5 ± 2.6 number of medications in average	Akner [4] found 6.6 \pm 3.2 in average		
SD: Standard deviation.			

Table 7. Summary of drug–drug interaction and clinical consequence.			
Drug-drug interaction	Clinical consequence		
Furosemide + Metoprolol	Hyperglycemia/hypertriglyceridemia		
Furosemide + Pantoprazole	Hypomagnesemia		
Furosemide + Sodium picosulfate	Fluid and electrolyte disturbances		
Metoprolol + Amlodipine	Reduction in HR, cardiac conduction and contraction		
Clopidogrel + Pantoprazole	\downarrow Bioactivation of clopidogrel \rightarrow No antiplatelet eff.		
HR: Heart rate.			

The mechanism behind DDIs between Furosemide and Metoprolol is not well known; however, studies suggest that both β -blockers and loop diuretics can alter the lipid profile by elevating levels of triglycerides [65,66]. β -Blockers may also contribute to the development of hyperglycemia by diminishing the release of insulin from pancreatic β -cells [65,67]. Based on the findings of this study, interaction between these two was on top of the list of the identified DDIs; therefore, caution should be taken when prescribing these two drugs. Further studies are required to elucidate the potential clinical consequences of using Furosemide and Metoprolol.

The second most common DDI identified in this study was between Furosemide and Pantoprazole. Pantoprazole and other proton-pump inhibitors have shown to cause hypomagnesemia [67]. Urgent clinical consequences of hypomagnesemia include tremors, convulsions, delirium and even coma [68]. Since hypomagnesemia is also a side effect of Furosemide [69], the coadministration of these two drugs without taking the necessary precautions to maintain a normal serum magnesium concentration could be problematic.

Furthermore, a DDI between Furosemide and Sodium picosulfate was identified. Sodium picosulfate should be administered with caution when combined with medications that can affect renal function (e.g., Furosemide), since it may result in an increased risk of fluid and electrolyte disturbances, which may lead to seizures, arrhythmias and prolonged QT interval [70]. Another identified potential DDI was Metoprolol and Amlodipine. The combined treatment with β-blockers and calcium channel blockers has additive effects on heart rate, cardiac conduction and cardiac contractility. Cases have been documented on concurrent treatment with these two classes of drugs leading to congestive heart failure, hypotension, bradycardia or even cardiogenic shock [71,72]. We identified that potential DDIs were common when Clopidogrel was used with Pantoprazole, and when Clopidogrel was used with Atorvastatin. The conversion of Clopidogrel into its active metabolite by CYP450 isoenzymes in the liver is necessary for Clopidogrel to exert its inhibitory effect on platelet aggregation [73]. Pantoprazole, and other proton-pump inhibitors have been demonstrated to inhibit CYP450 2C9 and 2C19 in vitro [74], while Atorvastatin inhibits CYP450 34A [73]. Since CYP450 2C19 and 3A4 play a major role in Clopidogrel bioactivation, the coadministration of these drugs can reduce the activation and subsequent antiplatelet effect of Clopidogrel [73]. One study by Juurlink et al. [75] has shown an increase in the risk of recurrent myocardial infarction when treating with proton-pump inhibitors and Clopidogrel, simultaneously. However, this was not the case for Pantoprazole; hence, the clinical significance is yet to be fully established. Nonetheless, Doan et al. [29] observed that 80% of the hospitalized elderly population with polypharmacy had a potential CYP450-mediated DDI, thus, it is essential to be mindful when prescribing multiple drugs that interact with CYP450.

In this study, three of the ten most frequent DDIs included an antidiabetic drug. These DDIs included Metformin and Insulin, Metformin and Furosemide, and Insulin and Furosemide. Both Metformin and Insulin can lead to hypoglycemia [76], which is the most common side effect of glucose-lowering drugs in general [76].

Metformin is regarded as a low-risk drug for hypoglycemia, since it exerts its effect through a glucose-dependent mechanism [76]. However, several studies have observed an evident association between the use of insulin and higher risk of hypoglycemia [76]. A hypoglycemic episode leads to an activation of the autonomic nervous system resulting in minor symptoms, from sweating and palpitations, to more severe symptoms such as cognitive dysfunction and seizures. Depending on the severity and duration of the episode, hypoglycemia can even result in coma or death [76]. Considering these possible detrimental side effects of glucose-lowering drugs, in particular insulin, it is important to cautiously manage the treatment of a patient whose conditions require both types of drugs.

When using Furosemide in combination with Metformin, Furosemide increases the plasma concentration level of Metformin by 22% [77]. The clinical implications of this increase are uncertain [77], but it may be possible that it can result in lactic acidosis, which is a known reaction to Metformin [78]. When the plasma concentration of Metformin increases, the risk of this reaction could also increase.

The combination of Furosemide and Insulin can exert an inhibitory effect on the secretion of Insulin from the β -cells of the pancreas, which can generate a problem for patients receiving Insulin as medication as the patient's dose takes endogenous insulin production into account. This can result in hyperglycemia, due to the overall lower amount of insulin circulating in the blood. To account for this interaction, a dose adjustment should be considered [79].

Since several of the most common DDIs included antidiabetic drugs, it should also be noted that people with Type 2 diabetes mellitus are often of advanced age [80] and at a higher risk of polypharmacy, which makes them more exposed to potential DDIs and side effects [81].

A potential DDI between Metoprolol and Mirtazapine was identified. Newer antidepressants are known to have cardiovascular side effects [82]. As such, coadministration of Mirtazapine with the β-blocker Metoprolol would presumably lead to additive hypotensive effects, as well as changes in pulse and/or heart rate. However, it has been observed that coadministration of Metoprolol and Mirtazapine is safe with no clinically significant DDIs [83]. In fact, this lack of clinical adverse effects has been documented to such a degree that Mirtazapine has been used as the negative control in a case–control study of interactions between Metoprolol and selective serotonin reuptake inhibitors [84].

The most common, potentially clinically relevant DDIs identified in this study were of moderate severity; however, major potential DDIs also occurred. Three of them consisted of an interaction between opioids and benzodiazepines (Codeine and Oxazepam, Codeine and Zolpidem, and Morphine and Oxazepam). The coadministration of opioids and benzodiazepines is associated with an increased risk of overdose deaths and emergency department visits [85]. Benzodiazepines have also been demonstrated to exert an exacerbating effect on opioid-mediated respiratory depression [85]. Additionally, a study has noted an elevated fracture risk among older men who are treated with a combination of opioids and benzodiazepines [86]. The FDA recommends that healthcare professionals minimize the prescription of opioid medications with benzodiazepines and other CNS depressants, when possible.

Strengths & limitations of the present study

Study population

This is a preliminary study with a relatively small-sample size (n = 100). Nevertheless, certain measures were taken to assure that the study population was as representative of the background population as possible. For instance, 5–20 randomized residents were identified by nurses from each of the eight nursing homes, instead of including a higher number of residents from only a few nursing homes. Furthermore, the participating nursing homes were of various geographical locations within Aalborg municipality. These approaches provided an advantage in terms of minimizing possible selection biases and confounders and enhance generalizability. However, in future studies with larger cohorts, considering both internal and external sources of bias must be taken into consideration. According to a census conducted by Aalborg municipality in 2019, 1860 nursing home residencies are currently available in Aalborg municipality. The prevalence of polypharmacy in the elderly has been estimated, by Scandinavian studies, range from as low as 16.8–30% within the range of 65–74 years to 33–65% at 95+ years [9,27]. Similar results have been highlighted by systematic review on the prevalence of polypharmacy by the WHO, where the estimated prevalence was 38.1–91.2% among nursing homes or long-term care facility residents. Our study did not aim at finding the prevalence but was a purely descriptive study. In future studies, sample size might be estimated if prevalence is the purpose of the study.

Lack of comparison

This study did not include a comparison to the rest of the nursing home residents, who received less than five medications. As far as the aim of this study was to elucidate the patterns and characteristics of polypharmacy, primarily related to DDIs, the inclusion of these residents with no polypharmacy did not seem relevant. This is because the amount of potential pairwise interactions increases exponentially with each coprescribed drug [63], making comparisons irrelevant. However, it is known that polypharmacy is associated with increased age, multimorbidity and female sex [3,27,50]. The lack of this comparison in the present study makes it impossible to determine whether the demographics of the study population differ significantly when compared with a nonpharmaceutical background population. For example, the female population of this study was 68%. It is difficult to conclude what this number reflects without a comparable group from the same population living in nursing homes in Denmark are females, it cannot be ruled out that a lower proportion of nursing home residents are females in Aalborg, with this study including these females to a higher degree.

Classification of drugs

By using the ATC codes, promoted by the European Medicines Agency to categorize medications, this study has tried to utilize a reliable system of drug classification in order to improve reproducibility and transparency. However, due to the lack of access to information regarding the general health status of the subjects, the categorization of drugs into therapeutic subgroups was complicated. Because a single drug can have different ATC codes depending on the therapeutic purpose [46] and lack of knowledge surrounding current health problems might have led to the condition that some drugs may have been categorized with a bias.

Definition of polypharmacy

Only prescription medications were included in this study, even though herbal and dietary supplements (HDS) can interact with prescription-based drugs [87,88]. Food supplements are defined as 'foodstuffs the purpose of which is to supplement the normal diet and which are concentrated sources of nutrients or other substances with a nutritional or physiological effect, alone or in combination, marketed in dose form' (The European Commission [89]). Because of this broad definition of food supplements alone, and the subsequent difficulty of defining and including all potentially relevant HDS, it was decided that HDS would be kept out of the scope of this study. However, due to the importance and practical benefits of such studies, it is recommended for a future investigation. This is mainly because the use of HDS is on the rise [90]. Cases have even been reported of so-called 'polyherbacy,' implying similarity to those associated with polypharmacy. It has been demonstrated that HDS–drug interactions can lead to an increased risk of DDIs [17], it would be rational to consider that polypharmaceutical patients are also at a higher risk of HDS–drug interactions. This suggests that there are possibly some potentially clinically relevant HDS–drug interactions, it is relevant to include HDS in future studies to acquire a thorough understanding of potential problems related to polypharmacy in elderly.

Conclusion

The findings from this study suggest that potentially inappropriate polypharmacy exists among elderly nursing home residents in Aalborg municipality. A direct correlation was determined between the number of drugs and number of DDIs. A complex relationship between age and number of drugs became evident, where age was inversely correlated with number of drugs. A significant difference in number of drugs was observed between the different age groups, where the highest median number of drugs was determined in the age group of 75–84 years, whereas the lowest medians of concomitant drugs were seen in the oldest age groups. In line with this finding, significant differences in the number of potential DDIs were observed in the FDA database between the different age groups. Tendencies were observed with higher number of drugs and more potential DDIs in females compared with males. The clinical consequences of the ten most commonly observed potential DDIs in this study consisted of hyperglycemia, hypotriglyceridemia, hypomagnesemia, fluid and electrolyte disturbances.

Future perspective

Polypharmacy in the elderly is still under recognized. There is a need for awareness and identification of cases with inappropriate polypharmacy among elderly. Identification of elderly residents in nursing homes with this condition can help determine better prevention strategies to enhance the quality of life of these individuals. Our findings call for further investigation and development of guideline strategies. Establishing a detailed understanding of the patterns and characteristics of potential inappropriate polypharmacy in the elderly could provide a foundation for minimizing health and economic consequences caused by inappropriate polypharmacy. To achieve this, a larger scale study is proposed – collecting sample data from different regions in Denmark and including a control group, the inclusion of factors such as diet, dietary supplements and consumption of alcohol and tobacco would also yield further information. Inclusion of medical records would make it possible to determine the incidence rate of the potential DDIs leading to actual clinical consequences. As it has also been highlighted by multiple Danish studies on polypharmacy, the current state of polypharmacy in Denmark calls for more thorough medication reviews in the elderly [92,93], for example, through introduction of the STOPP criteria [38–40] on a broader scale.

Summary points

- Polypharmacy is the concurrent use of more than one drug. Multimorbidity is common among the elderly and a leading factor of polypharmacy and its risks. Since the elderly population has grown tremendously, the frequency of polypharmacy can also be expected to increase. Thus, it is important to investigate this phenomenon to prevent inappropriate polypharmacy.
- Based on the current literature, it is difficult to assess how widespread inappropriate polypharmacy is in Denmark. Aim
- To identify patterns and characteristics of polypharmacy in the elderly living in nursing homes in Aalborg municipality, Denmark, by investigating the drug–drug interactions (DDIs) and potential adverse effects.

Methods

- Nursing homes in Aalborg municipality (n = 25) were contacted in November 2019 and asked to supply information from 20 randomly selected anonymized residents, among those aged 65 years or older, who were concurrently taking five or more medications. The collected information consisted of age, gender, co-morbidities and medication list.
- Two different drug interaction databases (one run by the Danish Medication Agency and the other by the US FDA) were utilized to identify potential DDIs among the medications taken by the residents and to divide DDIs into three levels of severity based on clinical significance.
- Descriptive statistics were applied.

Results

- The total number of nursing home residents included in this study was 100 (68% females; 32% males; mean age \pm standard deviation [SD]: 82.9 \pm 7.7 years).
- Five hundred and eighty-seven (mean \pm SD: 5.8 \pm 2.2) co-morbidities were registered with the cardiovascular system as the most commonly affected system. The total number of medications was 848 (mean \pm SD: 8.5 \pm 2.6) with the most common being medications related to the gastrointestinal system and metabolism.
- Based on the FDA database, the majority of observed potential interactions were of moderate severity (n = 786, 76.8%), whereas the Danish database identified the majority as being of mild severity (n = 226, 68.6%).
- Statistically significant differences in the number of drugs (p = 0.013) and number of DDIs per resident in the FDA database (p = 0.039) were observed among the four age groups (65–74, 75–84, 85–94, 95+). No sex-related difference was found.
- The number of drugs and DDIs were positively correlated.

Discussion & conclusion

- The findings suggest that potentially inappropriate polypharmacy exists among the studied elderly residents.
- Further research is needed to find if this small scope study could similarly reflect characteristics of polypharmacy in Denmark.
- Strategies and guidelines to prevent inappropriate polypharmacy in nursing homes are warranted.

Author contributions

The study idea was initiated by P Gazerani and discussed with J Astorp, M Gjela, P Jensen and RD Bak. The study was designed by all the authors, and conducted by J Astorp, M Gjela, P Jensen and RD Bak. Results were discussed and interpreted by all the authors. All the authors contributed in drafting the manuscript and agreed on the final format for submission.

Acknowledgments

The authors are grateful to the nursing homes Drachmannshave, Elmely, Hassinghave, Hellashoej, Sirius, Smedegaarden, Solgaarden and Storvorde. Without timely and efficient cooperation of these facilities, it was impossible to achieve the goals of this study.

Financial & competing interests disclosure

The authors have no relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript. This includes employment, consultancies, honoraria, stock ownership or options, expert testimony, grants or patents received or pending, or royalties.

No writing assistance was utilized in the production of this manuscript.

Ethical conduct of research

Due to the nature of this study, the ethical approval by the North Denmark Region Committee on Health Research Ethics was not required. Since participants' information was provided anonymously by nursing homes, no consent form was required.

Open access

This work is licensed under the Creative Commons Attribution 4.0 License. To view a copy of this license, visit http://creativecommons.org/licenses/by/4.0/

References

Papers of special note have been highlighted as: • of interest

- 1. Department of Economic and Social Affairs, United Nations. World population prospects: the 2019 revision (2019). www.un.org/development/desa/publications/world-population-prospects-2019-highlights.html
- Johnston MC, Crilly M, Black C, Prescott GJ, Mercer SW. Defining and measuring multimorbidity: a systematic review of systematic reviews. *Eur. J. Public Health* 29(1), 182–189 (2019).
- 3. Violan C, Foguet-Boreu Q, Flores-Mateo G *et al.* Prevalence, determinants and patterns of multimorbidity in primary care: a systematic review of observational studies. *PLoS ONE* 9(7), e102149 (2014).
- Akner G. Analysis of multimorbidity in individual elderly nursing home residents. Development of a multimorbidity matrix. Arch. Gerontol. Geriatr. 49(3), 413–419 (2009).
- Moth G, Vestergaard M, Vedsted P. Chronic care management in Danish general practice a cross-sectional study of workload and multimorbidity. *BMC Fam. Pract.* 13(52), doi: 10.1186/1471-2296-13-52 (2012).
- Straand J, Rokstad KS. Elderly patients in general practice: diagnoses, drugs and inappropriate prescriptions. A report from the More & Romsdal prescription study. *Fam. Pract.* 16(4), 380–388 (1999).
- Fano V, Chini F, Pezzotti P, Bontempi K. Estimating the prevalence and the determinants of polypharmacy using data from a health administrative database: a comparison of results obtained employing different algorithms. *Adv. Pharmacoepidemiol. Drug. Saf.* 3(2), DOI: 10.4172/2167-1052.1000151 (2014).
- Masnoon N, Shakib S, Kalisch-Ellett L, Caughey GE. What is polypharmacy? A systematic review of definitions. BMC Geriatr. 17(1), 230 (2017).
- It shows that polypharmacy is a public health issue with increasing prevalence and this is sensitive to applied algorithms used in this study.
- 9. Morin L, Johnell K, Laroche ML, Fastbom J, Wastesson JW. The epidemiology of polypharmacy in older adults: register-based prospective cohort study. *Clin. Epidemiol.* 10, 289–298 (2018).
- It presents the value of prospective and registered based study in the field.
- Gnjidic D, Hilmer SN, Blyth FM et al. Polypharmacy cutoff and outcomes: five or more medicines were used to identify community-dwelling older men at risk of different adverse outcomes. J. Clin. Epidemiol. 65(9), 989–995 (2012).
- Calderon-Larranaga A, Gimeno-Feliu LA, Gonzalez-Rubio F *et al.* Polypharmacy patterns: unravelling systematic associations between prescribed medications. *PLoS ONE* 8(12), (2013). https://journals.plos.org/plosone/article/file?id=10.1371/journal.pone.0084967&ctype=printable
- 12. World Health Organization. Medication safety in polypharmacy. WHO/UHC/SDS/2019.11 (2019). https: //apps.who.int/iris/bitstream/handle/10665/325454/WHO-UHC-SDS-2019.11-eng.pdf?ua=1
- 13. Soler O, Barreto JOM. Community-level pharmaceutical interventions to reduce the risks of polypharmacy in the elderly: overview of systematic reviews and economic evaluations. *Front. Pharmacol.* 10, 302 (2019).
- 14. Rodrigues MCS, De Oliveira C. Drug–drug interactions and adverse drug reactions in polypharmacy among older adults: an integrative review. *Rev. Lat. Am. Enfermagem* 24, e2800 (2016).

• This review provides an overview on drug-drug interactions in the field.

- 15. Nyborg G, Brekke M, Straand J, Gjelstad S, Romoren M. Potentially inappropriate medication use in nursing homes: an observational study using the NORGEP-NH criteria. *BMC Geriatr.* 17(1), 220 (2017).
- Bourgeois FT, Shannon MW, Valim C, Mandl KD. Adverse drug events in the outpatient setting: an 11-year national analysis. *Pharmacoepidemiol. Drug Saf.* 19(9), 901–910 (2010).
- 17. Gaeta TJ, Fiorini M, Ender K, Bove J, Diaz J. Potential drug–drug interactions in elderly patients presenting with syncope. *J. Emerg. Med.* 22(2), 159–162 (2002).
- 18. Maher RL, Hanlon J, Hajjar ER. Clinical consequences of polypharmacy in elderly. Expert Opin. Drug Saf. 13(1), 57-65 (2014).
- This expert opinion provides the importance of polypharmacy from consenquences point of view within the field.
- 19. Edwards IR, Aronson JK. Adverse drug reactions: definitions, diagnosis, and management. Lancet 356(9237), 1255-1259 (2000).
- Hanlon JT, Pieper CF, Hajjar ER et al. Incidence and predictors of all and preventable adverse drug reactions in frail elderly persons after hospital stay. J. Gerontol. Biol. 61(5), 511–515 (2006).
- 21. Lesher E. Validation of the geriatric depression scale among nursing home residents. Clin. Gerontol. 4(4), 21-28 (1986).
- 22. Wierenga PC, Buurman BM, Parlevliet JL *et al.* Association between acute geriatric syndromes and medication-related hospital admissions. *Drugs Aging* 29(8), 691–699 (2012).
- 23. Marcum ZA, Amuan ME, Hanlon JT *et al.* Prevalence of unplanned hospitalizations caused by adverse drug reactions in older veterans. *J. Am. Geriatr. Soc.* 60(1), 34–41 (2012).
- 24. Payne RA, Avery AJ. Polypharmacy: one of the greatest prescribing challenges in general practice. Br. J. Gen. Pract. 61(583), 83-84 (2011).
- 25. Klotz U. Pharmacokinetics and drug metabolism in the elderly. Drug Metab. Rev. 41(2), 67-76 (2009).
- 26. Mallet L, Spinewine A, Huang A. The challenge of managing drug interactions in elderly people. Lancet 370(9582), 185–191 (2007).
- Wastesson JW, Canudas-Romo V, Lindahl-Jacobsen R, Johnell K. Remaining life expectancy with and without polypharmacy: a register-based study of swedes aged 65 years and older. J. Am. Med. Dir. Assoc. 17(1), 31–35 (2016).
- 28. Kinirons MT, O'mahony MS. Drug metabolism and ageing. Br. J. Clin. Pharmacol. 57(5), 540-544 (2004).
- 29. Doan J, Zakrzewski-Jakubiak H, Roy J, Turgeon J, Tannenbaum C. Prevalence and risk of potential cytochrome P450-mediated drug–drug interactions in older hospitalized patients with polypharmacy. *Ann. Pharmacother.* 47(3), 324–332 (2013).
- 30. Davies EA, O'mahony MS. Adverse drug reactions in special populations the elderly. Br. J. Clin. Pharmacol. 80(4), 796-807 (2015).
- 31. Rambhade S, Chakarborty A, Shrivastava A, Patil UK, Rambhade A. A survey on polypharmacy and use of inappropriate medications. *Toxicol. Int.* 19(1), 68–73 (2012).
- 32. Dookeeram D, Bidaisee S, Paul JF *et al.* Polypharmacy and potential drug–drug interactions in emergency department patients in the Caribbean. *Int. J. Clin. Pharm.* 39(5), 1119–1127 (2017).
- Akazawa M, Imai H, Igarashi A, Tsutani K. Potentially inappropriate medication use in elderly Japanese patients. Am. J. Geriatr. Pharmacother. 8(2), 146–160 (2010).
- Hovstadius B, Astrand B, Persson U, Petersson G. Acquisition cost of dispensed drugs in individuals with multiple medications. A register-based study in Sweden. *Health Policy* 101(2), 153–161 (2011).
- Hovstadius B, Petersson G. The impact of increasing polypharmacy on prescribed drug expenditure-a register-based study in Sweden 2005–2009. *Health Policy* 109(2), 166–174 (2013).
- 36. Wang-Hansen MS, Wyller TB, Hvidsten LT, Kersten H. Can screening tools for potentially inappropriate prescriptions in older adults prevent serious adverse drug events? *Eur. J. Clin. Pharmacol.* 75(5), 627–637 (2019).

• It presents the value of screening tools.

- 37. Rognstad S, Brekke M, Fetveit A, Spigset O, Wyller TB, Straand J. The Norwegian General Practice (NORGEP) criteria for assessing potentially inappropriate prescriptions to elderly patients. A modified Delphi study. *Scand. J. Prim. Health Care* 27(3), 153–159 (2009).
- Gallagher P, Ryan C, Byrne S, Kennedy J, O'mahony D. STOPP (Screening Tool of Older Person's Prescriptions) and START (Screening Tool to Alert doctors to Right Treatment). Consensus validation. Int. J. Clin. Pharmacol. Ther. 46(2), 72–83 (2008).
- O'Mahony D, O'Sullivan D, Byrne S, O'Connor MN, Ryan C, Gallagher P. STOPP/START criteria for potentially inappropriate prescribing in older people: version 2. Age. Ageing 44(2), 213–218 (2015).
- O'Mahony D, O'Sullivan D, Byrne S, O'Connor MN, Ryan C, Gallagher P. Corrigendum: STOPP/START criteria for potentially inappropriate prescribing in older people: version 2. Age. Ageing 47(3), 489 (2018).
- 41. Sunhed.Dk. Polyfarmaci hos ældre. laegehaandbogen (2014). www.sundhed.dk/sundhedsfaglig/laegehaandbogen/geriatri/tilstande-og-sygdomme/behandling/polyfarmaci-hos-aeldre/
- 42. Society DG. Dansk Selskab for Geriatri (2020). https://danskgerontologi.dk/

- Rosholm JU, Bjerrum L, Hallas J, Worm J, Gram LF. Polypharmacy and the risk of drug–drug interactions among Danish elderly a prescription database study. Dan. Med. Bull. 45(2), 210–213 (1998).
- 44. Barat I, Andreasen F, Damsgaard EMS. Drug therapy in the elderly: what doctors believe and patients actually do. *Br. J. Clin. Pharmacol.* 51(6), 615–622 (2001).
- 45. Kommune Aalborg Municipality (2020). www.aalborg.dk/
- Methodology WCCFDS. Guidelines for ATC classification and DDD assignment (2020). www.whocc.no/atc_ddd_index_and_guidelines/guidelines/
- 47. Danish Medicines Agency (2020). https://laegemiddelstyrelsen.dk/
- 48. United States Food and Drug Administration Drugs.Com (2020). www.drugs.com/
- 49. Danish Statistics (Danmark statistik) (2020). www.dst.dk/da/Statistik/statistikbanken
- 50. Kaufman DW, Kelly JP, Rosenberg L, Anderson TE, Mitchell AA. Recent patterns of medication use in the ambulatory adult population of the United States the Slone survey. J. Am. Med. Assoc. 287(3), 337–344 (2002).
- Onder G, Carpenter I, Finne-Soveri H et al. Assessment of nursing home residents in Europe: the Services and Health for Elderly in Long TERm care (SHELTER) study. BMC Health Serv. Res. 12, 5 (2012).
- 52. Gurwitz JH. The age/gender interface in geriatric pharmacotherapy. J. Womens Health 14(1), 68-72 (2005).
- 53. Ældreministeriet S-O. National undersøgelse af forholdene på plejecentre (2016). https://sum.dk/~/media/Filer%20-%20Publikationer _i_pdf/2016/National-undersøgelse-af-forholdene-paa-plejec/Undersøgelse-af-forholdene-paa-plejecentre.pdf
- 54. Holmes HM. Rational prescribing for patients with a reduced life expectancy. Clin. Pharmacol. Ther. 85(1), 103–107 (2009).
- 55. Venturini CD, Engroff P, Ely LS *et al.* Gender differences, polypharmacy, and potential pharmacological interactions in the elderly. *Clinics (Sao Paulo)* 66(11), 1867–1872 (2011).
- Suominen-Taipale AL, Martelin T, Koskinen S, Holmen J, Johnsen R. Gender differences in health care use among the elderly population in areas of Norway and Finland. A cross-sectional analysis based on the HUNT study and the FINRISK Senior Survey. *BMC Health Serv. Res.* 6, 110 (2006).
- 57. Gijsbers Van Wijk CM, Kolk AM, Van Den Bosch WJ, Van Den Hoogen HJ. Male and female morbidity in general practice: the nature of sex differences. *Soc. Sci. Med.* 35(5), 665–678 (1992).
- 58. EUROSTAT. Medicine use statistics (2014). https://ec.europa.eu/eurostat/statistics-explained/index.php/Medicine_use_statistics#Prescribed_medicines
- 59. Kheshti R, Aalipour M, Namazi S. A comparison of five common drug–drug interaction software programs regarding accuracy and comprehensiveness. J. Res. Pharm. Pract. 5(4), 257–263 (2016).
- 60. Kuperman GJ, Bobb A, Payne TH et al. Medication-related clinical decision support in computerized provider order entry systems: a review. J. Am. Med. Inform. Assoc. 14(1), 29–40 (2007).
- 61. Sonnerstam E, Sjolander M, Lovheim H, Gustafsson M. Clinically relevant drug-drug interactions among elderly people with dementia. *Eur. J. Clin. Pharmacol.* 74(10), 1351–1360 (2018).
- Goldberg RM, Mabee J, Chan L, Wong S. Drug–drug and drug–disease interactions in the ED: analysis of a high-risk population. Am. J. Emerg. Med. 14(5), 447–450 (1996).
- 63. Fung KW, Kapusnik-Uner J, Cunningham J, Higby-Baker S, Bodenreider O. Comparison of three commercial knowledge bases for detection of drug–drug interactions in clinical decision support. J. Am. Med. Inform. Assoc. 24(4), 806–812 (2017).
- 64. Ministry of Health. https://sum.dk/Puljer/Satspuljeaftaler/2016_2019.aspx
- 65. Vue MH, Setter SM. Drug-induced glucose alterations part 1: drug-induced hypoglycemia. Diabetes Spectr. 24(3), 171–177 (2011).
- Herink M, Ito MK. Medication induced changes in lipid and lipoproteins. In: *Endotext*. Feingold KR, Anawalt B, Boyce A *et al.* (Eds). MA, USA (2000). www.ncbi.nlm.nih.gov/books/NBK326739/
- 67. Broeren MaC, Geerdink EaM, Vader HL, Bake AWLV. Hypomagnesemia induced by several proton-pump inhibitors. *Ann. Intern. Med.* 151(10), 755–756 (2009).
- Pham PC, Pham PA, Pham SV, Pham PT, Pham PM, Pham PT. Hypomagnesemia: a clinical perspective. *Int. J. Nephrol. Renovasc. Dis.* 7, 219–230 (2014).
- 69. Ryan MP, Devane J, Ryan MF, Counihan TB. Effects of diuretics on the renal handling of magnesium. *Drugs* 28(Suppl. 1), 167–181 (1984).
- Dailymed. LABEL: CLENPIQ- sodium picosulfate, magnesium oxide, and anhydrous citric acid liquid (2020). https://dailymed.nlm.nih.gov/dailymed/drugInfo.cfm?setid=a40a02ad-93ca-4658-b413-eb4017917034
- 71. Henry M, Kay MM, Viccellio P. Cardiogenic-shock associated with calcium-channel and beta-blockers reversal with intravenous calcium–chloride. *Am. J. Emerg. Med.* 3(4), 334–336 (1985).
- 72. Strauss WE, Egan T, Mcintyre KM, Parisi AF. Combination therapy with diltiazem and propranolol precipitation of congestive heart-failure. *Clin. Cardiol.* 8(6), 363–366 (1985).

- 73. Lau WC, Gurbel PA. The drug-drug interaction between proton pump inhibitors and clopidogrel. *Can. Med. Assoc. J.* 180(7), 699–700 (2009).
- Li XQ, Andersson TB, Ahlstrom M, Weidolf L. Comparison of inhibitory effects of the proton pump-inhibiting drugs omeprazole, esomeprazole, lansoprazole, pantoprazole, and rabeprazole on human cytochrome P450 activities. *Drug Metab. Disposition* 32(8), 821–827 (2004).
- 75. Juurlink DN, Gomes T, Ko DT *et al.* A population-based study of the drug interaction between proton pump inhibitors and clopidogrel. *Can. Med. Assoc. J.* 180(7), 713–718 (2009).
- 76. Kalra S, Mukherjee JJ, Venkataraman S *et al.* Hypoglycemia: the neglected complication. *Indian J. Endocrinol. Metab.* 17(5), 819–834 (2013).
- 77. Rivero Viera Y, Auyanet Saavedra I, Guerra Rodriguez R, Esparza Martin N, Fernandez Granados S, Garcia Canton C. Metformin and diuretics. *Nefrologia* 39(5), 557–558 (2019).
- Chang CT, Chen YC, Fang JT, Huang CC. Metformin-associated lactic acidosis: case reports and literature review. J. Nephrol. 15(4), 398–402 (2002).
- 79. Sandstrom PE, Sehlin J. Furosemide reduces insulin release by inhibition of Cl- and Ca2+ fluxes in beta-cells. *Am. J. Physiol.* 255(5 Pt 1), E591–E596 (1988).
- May M, Schindler C. Clinically and pharmacologically relevant interactions of antidiabetic drugs. *Ther. Adv. Endocrinol. Metab.* 7(2), 69–83 (2016).
- 81. Kirkman MS, Briscoe VJ, Clark N et al. Diabetes in older adults: a consensus report. J. Am. Geriatr. Soc. 60(12), 2342-2356 (2012).
- Pacher P, Kecskemeti V. Cardiovascular side effects of new antidepressants and antipsychotics: new drugs, old concerns? *Curr. Pharm. Des.* 10(20), 2463–2475 (2004).
- Crewe HK, Lennard MS, Tucker GT, Woods FR, Haddock RE. The effect of selective serotonin re-uptake inhibitors on cytochrome P4502D6 (CYP2D6) activity in human liver microsomes. Br. J. Clin. Pharmacol. 34(3), 262–265 (1992).
- Bahar MA, Wang Y, Bos JHJ, Wilffert B, Hak E. Discontinuation and dose adjustment of metoprolol after metoprolol-paroxetine/fluoxetine co-prescription in Dutch elderly. *Pharmacoepidemiol. Drug Saf.* 27(6), 621–629 (2018).
- 85. Hwang CS, Kang EM, Kornegay CJ, Staffa JA, Jones CM, Mcaninch JK. Trends in the concomitant prescribing of opioids and benzodiazepines, 2002–2014. Am. J. Prev. Med. 51(2), 151–160 (2016).
- 86. Reisfield GM, Webster LR. Benzodiazepines in long-term opioid therapy. Pain Med. 14(10), 1441–1446 (2013).
- 87. Choi JH, Ko CM. Food and drug interactions. J. Lifestyle Med. 7(1), 1-9 (2017).
- 88. Asher GN, Corbett AH, Hawke RL. Common herbal dietary supplement-drug interactions. Am. Fam. Physician 96(2), 101-107 (2017).
- 89. European Commission. Food suppliments. https://ec.europa.eu/food/safety/labelling_nutrition/supplements_en
- Nisly NL, Gryzlak BM, Zimmerman MB, Wallace RB. Dietary supplement polypharmacy: an unrecognized public health problem? Evid. Based Complement. Alternat. Med. 7(1), 107–113 (2010).
- Levy I, Attias S, Ben-Arye E, Goldstein L, Schiff E. Adverse events associated with interactions with dietary and herbal supplements among inpatients. Br. J. Clin. Pharmacol. 83(4), 836–845 (2017).
- Schiotz ML, Frolich A, Jensen AK et al. Polypharmacy and medication deprescribing: a survey among multimorbid older adults in Denmark. *Pharmacol. Res. Perspect.* 6(6), e00431 (2018). www.ncbi.nlm.nih.gov/pmc/articles/PMC6198567/
- Schaefer K, Maerkedahl H, Birk HO, Henriksen LO. Polypharmacy in general practice. *Dan. Med. Bull.* 57(7), A4165 (2010). www.ncbi.nlm.nih.gov/pubmed/20591340